Author's response to reviews

Title: Lower Vitamin D Status is More Common among Saudi Adults with Type 1 Diabetes Mellitus than Non-Diabetics

Authors:

Nasser Al-Daghri (aldaghri2011@gmail.com)
Omar Al-Attas (omrattas@ksu.edu.sa)
Majed Alokail (msa85@yahoo.com)
Khalid Alkharfy (alkharfy@ksu.edu.sa)
Sobhy Yakout (sobhy.yakout@gmail.com)
Naji Aljohani (najii@hotmail.com)
Hanan Alfawaz (halfawaz@ksu.edu.sa)
Eman Sheshah (eman_shesha@hotmail.com)
Mohammad Alharbi (biomarkers@ksu.edu.sa)

Version: 2 Date: 9 June 2013

Author's response to reviews: see over
June 6, 2013

**Professor Natalie Pafitis**
Executive Editor
BMC Public Health

Dear Prof. Natalie Pafitis,

On behalf of the authors I would like to resubmit our original article entitled "Lower Vitamin D Status is More Common among Adult Saudi Adults with Type 1 Diabetes Mellitus than Non-Diabetics", by Al-Daghri and colleagues. I confirm that this paper is not being considered elsewhere. I also confirm that the authors have seen the final version of the paper. We have already addressed the comments from the preliminary assessment.

Best regards,

Prof. Nasser Al-Daghri

**Corresponding Author**
Prince Mutaib Chair for Biomarkers of Osteoporosis
King Saud University Riyadh 11451, Saudi Arabia
Tel No: 0096614675939
Fax No: 0096614675939
E-mail: aldaghri2011@gmail.com